Author's response to reviews

Title: Dietary intake in the Personalized Medicine Research Project: a resource for studies of gene-diet interaction

Authors:

Lacie Strobush (strobush.laci@students.uwlax.edu)
Richard Berg (berg.richard@mcrf.mfldclin.edu)
Deanna Cross (cross.deanna@mcrf.mfldclin.edu)
Wendy Foth (foth.wendy@mcrf.mfldclin.edu)
Terrie Kitchner (kitchner.terrie@mcrf.mfldclin.edu)
Laura Coleman (coleman.laura@mcrf.mfldclin.edu)
Catherine McCarty (mccarty.catherine@mcrf.mfldclin.edu)

Version: 6 Date: 26 January 2011

Author's response to reviews: see over
January 26, 2011

Nutrition Journal Editorial Team
editorial@nutritionj.com

Dear Nutrition Journal Editorial Team,

RE: MS: 3490774464349462
Dietary intake in the Personalized Medicine Research Project: a resource for studies of gene-diet interaction

We have addressed all major and minor formatting revisions as outlined below.

Major revisions
Tables: tables should be legible on a standard page. If you cannot re-orientate the table to fit within a page, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.
All tables have been formatted to fit within a page.

In addition, we notice table 4 is constructed by two separate table tools. Please merge this table so that it is one full table.
Table 4 has been merged so that it is one full table.

References: Please ensure all references are cited and appear in consecutive numerical order
We have verified that all references are cited in consecutive numerical order.

Minor revisions
Tables: Please remove the visible vertical lines from the tables
All vertical lines have been removed from the tables.

Tables: Please do not use ranges when listing tables. Please list all the tables, for example:
Tables 5, 6, 7, 8 and 9 not Table 5-9)
All tables have been listed separately in the text.
Figures: Please ensure your figure is cited as 'Figure 1', rather than 'the figure'. The Figure is cited as Figure 1 in the text.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
We need assistance with the cropping.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
A final read-through has been done and some minor changes made for readability.

Sincerely,

Catherine A. McCarty, PhD, MPH, RD
Senior Research Scientist, Center for Human Genetics